Author's response to reviews

Title: Sampling and coverage issues of telephone surveys used for collecting health information in Australia: results from a face-to-face survey from 1999 to 2008

Authors:

Eleonora Dal Grande (eleonora.dalgrande@health.sa.gov.au)
Anne W Taylor (anne.taylor@health.sa.gov.au)

Version: 6 Date: 24 August 2010

Author's response to reviews:

The following formatting changes have been made to the document:

All authors have read and approved the final manuscript.

Initials have been used to refer to each author in the author's contribution section.

Tables 2 and 3 each have been made into one complete table.

Tables: tables 2 and 3 have been divided into parts, please ensure that each of these is formatted as one complete table.